Author’s response to reviews

Title: Development of an eHealth information resource for family carers supporting a person receiving palliative care on the island of Ireland.

Authors:

David Scott (david.scott@qub.ac.uk)

Peter Hudson (p.hudson@unimelb.edu.au)

Karen Charnley (kcharnley@aiihpc.org)

Cathy Payne (cpayne@aiihpc.org)

Gareth Westcott (gareth@olh.ie)

Version: 3 Date: 09 Aug 2019

Author’s response to reviews:

Dear Stephen

Thank you for your email of 06/08/2019 detailing the minor revisions required to our paper "Development of an eHealth information resource for family carers supporting a person receiving palliative care on the island of Ireland".

I have now:

1. Added an 'ethics approval' and 'consent to participate' statement within the 'Declarations' section.

2. Added a 'Consent for Publication' section.

3. I have amended the authors contributions section to contain the data analysis and interpretations section.

4. I have amended the headings to comply with the journals guidelines.

5. I have changed the title of the incorrect table to ensure that it now reads 'Table 2'
6. I have uploaded a clean copy of the manuscript with all changes accepted.

Many thanks for your help in this process. Our study is slightly unusual in that it describes the development process we undertook in creating a website and often, the work did not fit neatly into journal categories that were developed for the reporting of 'pure' research studies. Can I think you, your staff and the reviewers for the patience they have shown during this process.

Many thanks

David